ŗ



# **Geophysical Research Letters**<sup>•</sup>

## **RESEARCH LETTER**

10.1029/2024GL113393

#### **Key Points:**

- Single model simulations of the preindustrial last millennium suggest periods of heightened volcanic activity produce forced Atlantic Multidecadal Variability-like signals
- No evidence exists for internally generated oscillatory behavior on multidecadal timescales in the North Atlantic
- The North Pacific Ocean, in contrast to the North Atlantic, shows considerable internal variability

#### Supporting Information:

Supporting Information may be found in the online version of this article.

#### Correspondence to:

A. Fernandez, ferna476@umn.edu

#### Citation:

Fernandez, A., Steinman, B. A., Mann, M. E., & Christiansen, S. A. (2025). Multidecadal temperature variability in the Community Earth System Model Last Millennium Ensemble. *Geophysical Research Letters*, 52, e2024GL113393. https://doi.org/10.1029/2024GL113393

Received 4 NOV 2024 Accepted 26 APR 2025

#### **Author Contributions:**

Conceptualization: Alejandro Fernandez, Byron A. Steinman, Michael E. Mann Data curation: Alejandro Fernandez Formal analysis: Alejandro Fernandez Funding acquisition: Byron A. Steinman, Michael E. Mann, Shannon A. Christiansen Investigation: Alejandro Fernandez Methodology: Alejandro Fernandez, Michael E. Mann Project administration: Byron A. Steinman, Michael E. Mann, Shannon A. Christiansen Resources: Byron A. Steinman Software: Alejandro Fernandez

© 2025. The Author(s). This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# Multidecadal Temperature Variability in the Community Earth System Model Last Millennium Ensemble

Alejandro Fernandez<sup>1</sup>, Byron A. Steinman<sup>2</sup>, Michael E. Mann<sup>3</sup>, and Shannon A. Christiansen<sup>3</sup>

<sup>1</sup>Department of Earth and Environmental Sciences, University of Minnesota, Minneapolis, MN, USA, <sup>2</sup>Department of Earth and Environmental Sciences and Large Lakes Observatory, University of Minnesota Duluth, Duluth, MN, USA, <sup>3</sup>Department of Earth and Environmental Science, University of Pennsylvania, Philadelphia, PA, USA

**Abstract** We investigate the characteristics of inter- and multidecadal temperature variability in the Community Earth System Model Last Millennium Ensemble through spatiotemporal spectral analysis of forced and internal fields. We find high spectral density in North Atlantic (NA) and global temperature that is concurrent with periods of high volcanic activity, suggesting a forced origin. There is no evidence in the ensemble of an internally-generated and time-persistent signal for Atlantic Multidecadal Variability (AMV), the dominant mode on those timescales. The spatial patterns of low-frequency variability indicate activity throughout the North Pacific, where signals persist to a greater extent after the forced signal is removed, in contrast with the NA where only the subpolar region associated with deep water formation is active. Subtropical and tropical NA regions are strongly associated with forced responses, suggesting the canonical AMV pattern is comprised of both internal and forced components, with the latter being the main driver.

**Plain Language Summary** Climate change results from a combination of externally forced changes and internal, random variability. We used a series of simulations of Earth's temperature evolution during the preindustrial last millennium (850–1850 CE) from a single model to investigate the Atlantic Multidecadal Variability (AMV), a purported mode of potentially oscillatory climate variability detected in various climate data sets and thought to be the most influential mode in multidecadal timescales. By determining the forced component of variability using the mean of all simulations, we explore the extent to which AMV is internally generated or externally forced and identify its spatial expression. Our results indicate that during periods of high volcanic activity, a multidecadal oscillation can be detected. However, this signal is not present in the internal fields, suggesting that the real-world AMV is externally forced. Our methods allow for the spatial pattern reconstruction of any detected signals globally, which leads us to find more evidence for internal variability in the North Pacific than in the Atlantic. These findings support growing evidence that AMV is a combination of externally forced and internally generated variability, in which the forced component is dominant, hindering the prospect of its predictability.

#### 1. Introduction

Characterizing multidecadal climate variability and its underlying physical mechanisms is one of the greatest challenges for robust assessment of uncertainty in climate model simulations (Deser et al., 2012; Hawkins & Sutton, 2009; Lehner et al., 2020). Central to this challenge has been the difficulty in identifying and predicting low-frequency internal variability in the North Atlantic (NA) region, which influences climatic conditions in the surrounding North American and Eurasian continents (Semenov et al., 2010; Sutton & Hodson, 2005), Atlantic hurricane activity (Goldenberg et al., 2001; Ting et al., 2019), the northeast Brazil and Sahel/Indian monsoons (Knight et al., 2006; R. Zhang & Delworth, 2006), and has been linked to the so-called warming hiatus of the early 21st Century (Li et al., 2020; Yang et al., 2020). Originally named the Atlantic Multidecadal Oscillation (AMO; Kerr, 2000) due to a purported narrowband pattern of oscillatory behavior (Delworth & Mann, 2000; Folland et al., 1986), the term Atlantic Multidecadal Variability (AMV) is now preferred to describe the dominant mode of sea surface temperature (SST) variability over the NA. This allows for a wider range of temporal and spatial expressions (R. Zhang, 2017), as well as implicitly including both internal oscillatory and non-oscillatory behavior and Atlantic surface temperature responses to external forcing (Frankignoul et al., 2017). As the dominant mode of NA and global climate variability in multidecadal timescales (Chylek et al., 2014), the AMV has also been observed to have an impact on the Pacific Decadal Oscillation (PDO) or Interdecadal Pacific Oscillation (IPO), through extratropical (e.g., R. Zhang & Delworth, 2007; L. Zhang & Zhao, 2015) or tropical



A. Steinman

Alejandro Fernandez Writing – review & editing:

A. Christiansen

Michael E. Mann, Shannon

Supervision: Byron A. Steinman, Michael E. Mann, Shannon A. Christiansen

Validation: Alejandro Fernandez, Byron

Alejandro Fernandez, Byron A. Steinman

Visualization: Alejandro Fernandez Writing – original draft: (e.g., Meehl et al., 2021; Ruprich-Robert et al., 2017) atmospheric teleconnections. The spatial and temporal expressions and climate impacts of AMV have been documented through studies of observations, proxy data and model simulations (e.g., Enfield et al., 2001; Michel et al., 2022; Wang et al., 2017), yet the specific mechanisms that underlie this pattern are still subject to debate (Bellucci et al., 2017; Delworth & Mann, 2000; Schlesinger & Ramankutty, 1994).

Although there is consensus that AMV is a combination of internally generated variability—related to the Atlantic Meridional Overturning Circulation (AMOC) and its responses to stochastic ocean-atmosphere interactions (Kim et al., 2020), or other modes of variability like the North Atlantic Oscillation (NAO; Delworth et al., 2017)—and SST responses to anthropogenic and natural forcings (Deser & Phillips, 2021; Qin et al., 2020), the extent to which internal versus external processes influence AMV is a matter of contention (Baek et al., 2022; Bellomo et al., 2018; Booth et al., 2012). A forced origin for AMV has been proposed by a number of studies wherein volcanic activity is identified as the driver of multidecadal SST variability (e.g., Birkel et al., 2018; Klavans et al., 2022; Otterå et al., 2010). Contrastingly, other groups have concluded that AMV is primarily internally driven through connections with AMOC and with a limited (or modulating) role for external forcing factors (e.g., Kim et al., 2020; Qin et al., 2020; Robson et al., 2015), coupling and amplifying of the NAO (Delworth et al., 2017; Sutton et al., 2018) and cloud-radiation feedbacks (Bellomo et al., 2016; Cane et al., 2017). In a series of efforts, Mann et al. analyzed control and forced CMIP5 simulations and suggested that AMV is predominantly a response to external forcing in the Last Millennium (Mann et al., 2021) and to anthropogenic aerosol forcing in the industrial era (Mann et al., 2020).

The characterization of AMV is complicated by the lack of consensus on its defining index-commonly calculated over the entire NA basin—which leads to inconsistent results (Robson et al., 2023). Furthermore, common methods for isolating internal AMV such as linear detrending of the low pass filtered and spatially averaged NA SST timeseries (i.e., the canonical AMV index; Enfield et al., 2001), subtracting global mean SST from the NA series (Trenberth & Shea, 2006), and other methods (e.g., Steinman et al., 2015; Ting et al., 2009) have been found insufficient to varying degrees (Deser & Phillips, 2021), and more sophisticated methods have been proposed (Deser & Phillips, 2023; Frankignoul et al., 2017; Wills et al., 2020). The range of methodologies for characterizing the internal AMV, compounded with the variety of time periods in both industrial and preindustrial data sets, have led to a range of periodicities from 20 to 40 years to as long as 60-80 years (Mann et al., 2020; Zhou et al., 2020). Observational data sets are likely too short to accurately encompass multidecadal variability (Mann et al., 2021) and are heavily influenced by anthropogenic GHG and aerosol forcing (Bellucci et al., 2017; Booth et al., 2012), while the estimation of internal variability in model simulations is known to be dependent on the choice of forcing series (Fyfe et al., 2021). When large ensembles of model simulations are available, however, the use of the single-model ensemble mean (Deser & Phillips, 2021; Mann et al., 2022), or of several models (Knight, 2009; Kravtsov & Callicutt, 2017), has been shown to adequately represent the forced signal (Frankcombe et al., 2018).

Here we characterize the spatiotemporal expressions of multidecadal variability in global surface temperature fields and explore their relationship to the canonical AMV, the main mode of variability over those timescales, by expanding on the methods and scope of Mann et al. (2020, 2021). We focus on the pre-industrial last millennium (850–1850 CE) to avoid two key issues: (a) the historical period is insufficiently long to capture more than a few cycles of a multidecadal signal, which makes the significance of any results tenuous, as the secular band is close to the upper range of AMV periodicity (Mann et al., 2020), and (b) the competing influences of anthropogenic GHG and aerosol emissions with natural forcings makes the isolation of internal signals challenging (Thompson et al., 2015). In contrast with Mann et al. (2020, 2021), we make use of a single model ensemble, thereby avoiding structural uncertainties present in multiple model comparisons and accurately determining the forced temperature signal (Frankcombe et al., 2018). Additionally, by employing Multi-taper Method Singular Value Decomposition (MTM-SVD) on the data fields, we are able to detect and reconstruct any pattern in the ensemble without calculating an AMV index and thus without any a-priori assumptions regarding its spatial and temporal characteristics. Furthermore, to explore the extent to which volcanic forcing impacts multidecadal SST variability, we also conduct spectral analyses across specific time periods of high and low volcanic forcing, by use of a moving window MTM-SVD approach, and compare resulting spectra in the NA to global fields under the assumption that changes in radiative forcing manifest themselves on a global scale while AMV-related variability would be strongest in the NA region. Lastly, we explore the global spatial patterns associated with purported oscillatory



behavior over a range of timescales to assess the consistency with which the patterns are expressed across the ensemble and compare forced versus internal expressions in the NA and Pacific regions.

#### 2. Data and Methods

#### 2.1. Data: The CESM LME

The Community Earth System Model (CESM) Last Millennium Ensemble (LME; Otto-Bliesner et al., 2016) is the largest available set of single model simulations spanning the past millennium, thereby providing a unique testbed for exploring climate variability across multiple timescales. We focus on the "All Forcing" ensemble (henceforth "LME"; N = 13), forced by volcanic, solar, orbital, GHG, aerosol and land-use changes (see Otto-Bliesner et al., 2016) and estimate the forced signal as the ensemble mean ("LME-EM") which is then subtracted from each individual ensemble member to isolate the internal variability ("LME-unforced"; Frankcombe et al., 2018). The volcanic forcing series used in the LME (Gao et al., 2008) is also used in CMIP5/PMIP3 last millennium protocols (Bothe et al., 2013), allowing for direct comparison between our results and previous analyses of those data (Mann et al., 2021). The LME consists of fully coupled transient simulations using version 1.1 of CESM with the CAM5 atmospheric model (Hurrell et al., 2008), land and atmosphere resolution of ~2° and ocean and sea ice of ~1°. We analyze surface air temperature fields which are strongly correlated with SSTs and yield nearly identical results for ocean regions.

#### 2.2. Method: Spectral Analysis of Climate Timeseries and Fields

We perform power-spectral-density analysis (PSD) via the Multi-taper Method (MTM; Thomson, 1982) on global and NA (defined as the region bound by 80°W-0°E) area-weighted average temperature series to detect broad spatial-scale signals and for the sake of comparison with similar studies (Ba et al., 2014; Bellomo et al., 2018; Miao et al., 2024). Additionally, we apply wavelet analysis (Torrence & Compo, 1998) to LME-EM, allowing for visualization of the spectral density in individual timeseries as it changes through time. Due to the limitations in detecting periodic signals that may weakly project onto global or regional averages (like AMV), we employ the MTM-SVD method, which was designed for the detection of narrowband spatiotemporal signals in climate data fields (Mann & Park, 1994, 1999). Unlike data decompositions in the time domain such as PCA/ EOF, MTM-SVD performs PSD analysis on each timeseries, transforming the data into the frequency domain before detecting patterns of variability through SVD. The process is done over a bandwidth  $\pm p_f$ , where  $f_r$  is the minimum resolvable frequency ( $f_r = 1/N \Delta t$ ; N samples  $\Delta t$  timestep) and p is a bandwidth parameter determined by p = K - 1, with K being the number of tapers in MTM. The Local Fractional Variance (LFV) is the signal detection variable whose magnitude indicates the relative level of spatiotemporal activity in a specific frequency band. Under the assumption that a red noise spectral background varies modestly in the narrow bandwidth of the analysis and thus locally approximates white noise, significance limits are calculated through bootstrapping wherein LFV spectra are calculated for temporally randomized fields. Once a signal is identified, it may be reconstructed from the complex SVD decomposition. Percent-variance-explained maps of the reconstructed signal versus the original fields are calculated to construct the spatial patterns. See Supporting Information S1 for a full description of the method.

#### 3. Results and Discussion

We first apply PSD and wavelet analysis to investigate signals that may have a significant imprint on North Atlantic mean surface temperatures and that are persistent throughout the length of the simulations. The timeseries analyzed here are analogous to the canonical AMV index (area-weighted mean surface temperatures; Figure 1a Top). MTM analysis on each LME and LME-unforced ensemble member characterizes the total variability and internal-only components of the ensemble, while LME-EM corresponds to the forced-only signal (Figure 1b Right). Wavelet analysis of LME-EM (Figure 1b, Left) represents the temporal evolution of spectral power of the forced signal and is directly related to the MTM spectrum of LME-EM. Two distinct periods of enhanced forced activity, 1200–1300 CE and 1600–1850 CE, both due to high rates of volcanism, are evident in the wavelet analysis. Note that these periods show heightened activity across a wide range of periodicities but are temporally bounded by relative volcanic quiescence. MTM spectra of the LME ensemble and LME-EM show enhanced power in the ~40- and ~60-year periodicities, which is absent in the unforced data. Indeed, diminished spectral density through all frequencies is observed in the LME-unforced series, suggesting that through the last



# **Geophysical Research Letters**



Figure 1. (a) Top: LME Ensemble Mean (LME-EM) North Atlantic mean surface air temperature (NAMSAT) anomaly timeseries (black line) and spread of the LME ensemble (IQR and 95% spread in dark and light red, respectively). All timeseries are 20-year lowpass filtered only for visualization. Bottom: Gao et al. (2008) global stratospheric volcanic aerosol injection standardized series (z-scored; red) and 100-year trailing average of the same data (dark gray) to highlight centennial-scale fluctuations. (b) Left: Continuous wavelet transform of the LME-EM. Light gray outlines represent 90% confidence levels. Right: MTM spectrum mean (solid) and IQR (shading) for LME, LME-unforced and LME-EM.

millennium a large proportion of oscillatory or pseudo-oscillatory behavior in mean NA SST (and thus AMV index) is driven by external forcing. Figure S1 in Supporting Information S1 shows the same analysis applied to global temperature series, wherein similar spectral features are present in the volcanically active periods of the last millennium, further suggesting that volcanic forcing is the main driver of multidecadal variability.

Studies comparing output from CESM and other models, paleoclimate proxy data and reanalysis products have reported enhanced spectral density in the  $\sim$ 60–80-year band for NA SSTs for the second half of the last millennium (post 1250 CE), but not before then (Dai et al., 2022; Mann et al., 2021; Miao et al., 2024). Our results suggest that these observations are likely due to high volcanic activity during the 13th, 17th, and 18th Centuries that manifests over a broad band of frequencies. Importantly, a comparison between the MTM spectra and the wavelet reveals that spectral peaks in certain frequencies do not correspond to signals consistently expressed through time, but are localized temporally near the high-volcanism periods. This suggests that multidecadal variability attributed to the AMV in PSD analyses may be the result of time-varying forcing mechanisms whose



### **Geophysical Research Letters**



**Figure 2.** Left and Middle: Moving-window (200-year) LFV spectra of North Atlantic and global temperature fields from the LME and LME-unforced. Dashed, thin solid and thick solid contours represent 50%, 90%, and 95% confidence levels, respectively. Thick horizontal ranges denote the periods of analysis for subsequent signal reconstruction (Figure 3). Right: LFV spectra of NA and global temperature fields for LME (red) and LME-unforced (blue). Mean and 2σ range of spectra represented by thick lines and shading.

impacts in distinct periods may be sufficient to produce to a spectral peak for the full series. In analyzing the mean spectrum of all simulations, as well as the spread of individual spectra, we find no common periodic behavior in the CESM LME NA mean surface temperature series that points to a mode of internal variability expressed on the timescales of interest. This indicates that CESM, in contrast to other models, does not produce significant internal, multidecadal variability in the North Atlantic (Ba et al., 2014; Fang et al., 2021; Mavilia et al., 2018).

The internal portion of the AMV is thought to be a redistribution of heat over the NA, and it may not necessarily project strongly onto mean temperature series (Mann et al., 2021) such as the AMV index or global series. To explore the spectral power of the temperature fields as a whole, we perform MTM-SVD analysis (Mann & Park, 1994) to spatially and temporally characterize any prevalent signal. By analyzing both NA and global fields we are able to discern if an AMV-like pattern is due to global variability (e.g., cooling due to volcanism) or localized phenomena (ocean circulation). Though the mean LFV spectra of the simulations (Figure 2, Right) show no statistically significant narrow-band signal through the last millennium, a moving-window MTM-SVD analysis reveals that apparent oscillatory behavior in the temperature fields is detected in all realizations only during times of strong forcing (Figure 2, Left). Signals above the 90% confidence level are absent in the NA analysis except for short ranges of time within ~100 years of the Samalas eruption of 1,257 CE, the largest eruption in the Gao et al. (2008) volcanic forcing record, and the global analysis shows highly significant signals close to the secular band (100 years) at the same time. A similar analysis of a much smaller (N = 5) ensemble of experiments forced only by volcanism shows similar trends in all time periods and spectral bands (Figure S2 in Supporting Information S1), further indicating that volcanism drives any apparent periodic behavior in CESM. Once internal variability is isolated, no spatiotemporally coherent signal exists that is common to all experiments

and that can be distinguished from a colored noise null hypothesis (Figure 2 Middle). Time-slice comparison of LFV spectra in the LME and volcanic-only simulations produces similar results, wherein discernible differences between the forced and unforced fields are only present in periods of high volcanism (Figure S2 in Supporting Information S1).

After assessing the level of spectral power across different time periods in the CESM LME, we further leverage MTM-SVD to characterize the spatial patterns associated with prominent signals. We analyze temperature fields during the period of highest LFV spectral density related to heightened volcanic activity (~1150–1400 CE) and calculate mean %-variance-explained maps that show the dominant spatial patterns in the ensemble. As no spatial regressions or correlations are necessary to characterize spatial patterns of variability in a particular frequency slice, no assumptions about the relationship between forced and unforced signals with any indexes or features in the data are needed (Mann et al., 2021), providing an unbiased view of the locations where significant temperature variability occurs. Furthermore, because the MTM-SVD method is capable of reconstructing a signal regardless of its relative amplitude or spatial extent, we are able to produce global maps of variance explained that provide context as to other manifestations of low-frequency temperature variability outside of the NA, particularly in the Pacific, and that do not dampen signals exclusive to the NA (Figure S4 in Supporting Information S1). We focus on bidecadal (~23 years), interdecadal (~44 years), multidecadal (~60 years) and centennial (~95 years) periodicities (Figure 3), chosen with the goal of capturing the highest LFV values in the evolutive spectra (Figure 2). Though the multidecadal slice does not show heightened spectral density in the MTM-SVD analysis, we show it due to its prevalence in previous works (Dai et al., 2022; Lapointe et al., 2020; Mann et al., 2021; Miao et al., 2024).

The signal reconstruction and resulting variance-explained maps of the forced ensemble (Figure 3 Left) display general similarities across the selected timescales such as heightened activity in the Gulf of Mexico, tropical and subtropical NA and subtropical Pacific. Previous studies characterizing spatial patterns of variability due to volcanism show similar spatial distributions on decadal and multidecadal timescales (Bellomo et al., 2018; Deser & Phillips, 2021; Fang et al., 2021; Watanabe & Tatebe, 2019). As the global impacts of volcanic cooling in CESM have been shown to strongly influence tropical oceans (Otto-Bliesner et al., 2016), observed activity in the tropical and southern Atlantic and Indian Oceans are expected in the forced fields. The NA activity seems to disappear, or attenuate significantly, in the internal fields (Figure 3, Right), suggesting that it is not predominantly the result of the model's reproduction of internal variability in the form of oceanic circulation or interactions with atmospheric phenomena. An exception is the subpolar NA region where the internal signal is relatively strong, likely because of internal variability related to AMOC and other ocean-atmosphere circulation patterns like the NAO (Frankignoul et al., 2017; McCarthy et al., 2015).

Although volcanic eruptions typically have a short-term direct effect on polar NA SSTs through sea ice expansion, when clustered they can have longer lasting impacts due to the ice-albedo feedback, which ultimately results in freshwater injection that lowers salinity and slows AMOC convection (Dai et al., 2022; Slawinska & Robock, 2018). In effect, it is suggested that a series of large eruptions in the late 13th Century, starting with the Samalas event, triggered the centuries-long cooling that led to the Little Ice Age (LIA; Miller et al., 2012). Our MTM-SVD analysis captures decadal and multidecadal processes in this subpolar region that are a combination of internal variability and long-term responses to volcanic forcing, though there appears to be no significant narrow-band periodicity to the signal. Thus, our results suggest that the long-lasting impacts of heightened volcanism during the 13th and 14th centuries leads to an amplification of NA SST variance by modulating AMOC activity through sea-ice expansion, reduced heat loss and freshwater injection, which would impact convection (Delworth & Mann, 2000). Periods in the last millennium with higher volcanic activity, as well as single large volcanic events, appear to have a greater AMOC variability on both annual and interdecadal timescales (Figure S5 in Supporting Information S1). Otto-Bliesner et al. (2016) similarly document how some eruptions have an immediate cooling influence on AMV followed by a decadal-scale warming, highlighting how volcanic effects influence surface temperatures and ocean circulation on different timescales.

The patterns of variability in the Pacific Ocean are more persistent between the LME and LME-unforced fields in contrast with the NA, with the strongest signal resembling the extratropical boundaries between the PDO dipoles (Hua et al., 2018; Takahashi & Watanabe, 2016; Xu et al., 2024). This could suggest that ocean circulation has a stronger influence on surface temperatures in the Pacific and therefore a weak or non-existent causal relationship between the AMV and IPO/PDO. This supports the results of Frankignoul et al. (2017), who suggest a larger





Figure 3. Mean spatial patterns of %-Variance Explained by reconstructed signals at each of the periocities identified in Figure 2 for LME (left) and LME-unforced (right) temperature fields across all 13 ensemble members during 1150–1400 CE.

proportion of Pacific variability to be internal compared to the AMV, and whose estimate of the PDO pattern is similar to our interdecadal and multidecadal results. The lack of signal concentrated in the tropical Pacific suggests that the impact of ENSO variability on lower frequency Pacific patterns is reflected at higher latitudes in the chosen timescales. Furthermore, the concentration of explained variance in the subtropics has a spatial pattern resembling that of the Pacific Meridional Mode (PMM; Amaya, 2019; Sanchez et al., 2019). Two regions of the Pacific show enhanced activity in the unforced fields: the western subtropical Pacific (off the coast of Japan), and the southern subtropical eastern Pacific (west Peru and northern Chile), regions usually associated with the PDO (Newman et al., 2016). The enhanced signal in the internal fields suggests CESM generates internal variability in the Pacific to a greater extent than in the NA.

#### 4. Conclusions

We employed the MTM-SVD method of spatiotemporal signal detection and reconstruction, coupled with single timeseries PSD and wavelet analysis, to assess evidence in the CESM LME for periodic or oscillatory signals in North Atlantic and global surface temperature fields, with a focus on the AMV. Our results build on those of Mann et al. (2020, 2021) by performing single-series spectral analysis on the AMV index used in other studies (Bellomo et al., 2018; Dai et al., 2022; Miao et al., 2024), performing signal reconstructions on both forced and unforced fields, implementing a moving window spectrum allowing for signal detection in distinct time intervals, and performing global signal reconstructions over a range of timescales and thereby comparing internal versus forced signals in both the NA and the Pacific. We found spectral peaks in the LME ensemble temperature fields that are absent in LME-unforced and that are exclusively present both in NA and global temperature series during time periods of high volcanic activity (Figure 1, Figure S1 in Supporting Information S1), agreeing with previous studies that suggest a strong role of forcing on AMV (e.g., Birkel et al., 2018; Clement et al., 2015; Otterå et al., 2010). Our results indicate that apparent oscillatory behavior in the NA is due to external forcing, specifically volcanic activity during the 13th and, to a lesser extent, the 17th and 18th centuries, as similar spectral behavior is evident in analyses of global fields (Figure 2), which indicates a common underlying mechanism to both. Previous studies comparing the spectral density in AMV indices prior to the Medieval Climate Anomaly and into the Little Ice Age report similar results, suggesting a shift from weak interdecadal to strong mulitidecadal periodicities of the AMV concurrent with the strengthening of volcanic forcing that led to the onset of the LIA (Dai et al., 2022; Miao et al., 2024). By identifying periods of high spectral density in multidecadal bands that are not persistent through time, and that are common to global and NA fields, as well as to both the LME and Volcanic-only experiments (Figure 2, Figures S2 and S3 in Supporting Information S1), we conclude that no internal mode of variability with any characteristic or predictable frequency can be detected in the climate as simulated by the CESM model.

After determining the spectral characteristics of the ensemble, we leveraged MTM-SVD signal reconstruction on global fields in order to disentangle the spatial patterns associated with any periodic behavior. Our results (Figure 3) agree with previous studies that have attributed a larger proportion of AMV variability to NA responses to external forcing than to internal climate dynamics (Frankignoul et al., 2017; Murphy et al., 2017; Watanabe & Tatebe, 2019) and show a spatial expression of variance explained that has commonalities with previous studies characterizing the AMV and/or Pacific SST variability (Knight et al., 2005; O'Reilly et al., 2019; Takahashi & Watanabe, 2016). The subpolar Atlantic region in particular shows evidence of internally generated temperature variability, while the rest of the canonical "horseshoe" shape often attributed to the AMV appears to be related to the Atlantic's response to radiative forcing, particularly in the tropical NA (Frankignoul et al., 2017). There is, however, no evidence that internal variability in the subpolar NA is oscillatory. Thus, we suggest that multi-decadal variability on timescales commonly attributed to the AMV is due to different mechanisms: an internally generated source of temperature variability (overprinted by external forcing) that occurs in the subpolar region of NA deep water formation but is not consistently periodic through the last millennium, and an exclusively forced region of temperature variability related to the response of the NA to volcanic activity, localized in the subtropical (~20–30°N) NA and Gulf of Mexico and that extends to the eastern branch of the canonical AMV pattern.

We further find that the spatial expressions of decadal and multidecadal patterns in the Pacific (IPO/PDO) are both stronger and more consistently expressed in the forced and unforced fields compared to those in the Atlantic, suggesting ocean and atmosphere dynamics exert a greater control on Pacific low-frequency variability in the LME. However, these observations may also be due to the long-term impact that volcanism has on the extent and characteristics of ENSO activity (McGregor et al., 2020; Pausata et al., 2023), which itself is strongly correlated with lower-frequency IPO variability in both observations (Heidemann et al., 2024) and model simulations (Capotondi et al., 2020).

The apparently strong dependence of inter- and multidecadal variability in NA SSTs to external forcing in CESM suggests that there is limited predictability for an AMV-like mode that could be used to reduce uncertainty in climate projections, though some skill has been found in initialized model experiments related to subpolar NA heat content and gyre circulation (Yeager, 2020). It has been shown, however, that CESM does not fully capture low frequency internal variability related to the AMV and NAO (O'Reilly et al., 2019) and that it is highly sensitive to volcanic radiative forcing (Otto-Bliesner et al., 2016). Thus, additional large simulation ensembles from different models, and from which internal variability could be isolated, would help elucidate the extent to



Acknowledgments

which the internal component of the AMV has changed through time and the degree to which multidecadal variability in the NA is a direct consequence of internal ocean-atmosphere dynamics versus external forcing.

#### **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

#### **Data Availability Statement**

All CESM LME simulations are publicly available and can be found at https://www.cesm.ucar.edu/communityprojects/lme. Code for MTM and MTM-SVD analysis of the data is publicly available as a Zenodo software release (Fernandez, 2024).

#### References

This work was funded by the National Science Foundation Paleoclimate Program (1748097, 1748115). Amaya, D. J. (2019). The Pacific meridional mode a 1007/s40641-019-00142-x

- Amaya, D. J. (2019). The Pacific meridional mode and ENSO: A review. *Current Climate Change Reports*, 5(4), 296–307. https://doi.org/10.1007/s40641-019-00142-x
- Ba, J., Keenlyside, N. S., Latif, M., Park, W., Ding, H., Lohmann, K., et al. (2014). A multi-model comparison of Atlantic multidecadal variability. *Climate Dynamics*, 43(9), 2333–2348. https://doi.org/10.1007/s00382-014-2056-1
- Baek, S. H., Kushnir, Y., Ting, M., Smerdon, J. E., & Lora, J. M. (2022). Regional signatures of forced North Atlantic SST variability: A limited role for aerosols and greenhouse gases. *Geophysical Research Letters*, 49(8), e2022GL097794. https://doi.org/10.1029/2022GL097794
- Bellomo, K., Clement, A. C., Murphy, L. N., Polvani, L. M., & Cane, M. A. (2016). New observational evidence for a positive cloud feedback that amplifies the Atlantic Multidecadal Oscillation. *Geophysical Research Letters*, 43(18), 9852–9859. https://doi.org/10.1002/2016GL069961
- Bellomo, K., Murphy, L. N., Cane, M. A., Clement, A. C., & Polvani, L. M. (2018). Historical forcings as main drivers of the Atlantic multidecadal variability in the CESM large ensemble. *Climate Dynamics*, 50(9–10), 3687–3698. https://doi.org/10.1007/s00382-017-3834-3
- Bellucci, A., Mariotti, A., & Gualdi, S. (2017). The role of forcings in the twentieth-century North Atlantic multidecadal variability: The 1940–75 North Atlantic cooling case study. *Journal of Climate*, *30*(18), 7317–7337. https://doi.org/10.1175/JCLI-D-16-0301.1
- Birkel, S. D., Mayewski, P. A., Maasch, K. A., Kurbatov, A. V., & Lyon, B. (2018). Evidence for a volcanic underpinning of the Atlantic multidecadal oscillation. Npj Climate and Atmospheric Science, 1(1), 1–7. https://doi.org/10.1038/s41612-018-0036-6
- Booth, B. B., Dunstone, N. J., Halloran, P. R., Andrews, T., & Bellouin, N. (2012). Aerosols implicated as a prime driver of twentieth-century North Atlantic climate variability. *Nature*, 484(7393), 228–232. https://doi.org/10.1038/nature10946
- Bothe, O., Jungclaus, J. H., & Zanchettin, D. (2013). Consistency of the multi-model CMIP5/PMIP3-past1000 ensemble. *Climate of the Past*, 9(6), 2471–2487. https://doi.org/10.5194/cp-9-2471-2013
- Cane, M. A., Clement, A. C., Murphy, L. N., & Bellomo, K. (2017). Low-pass filtering, heat flux, and Atlantic multidecadal variability. Journal of Climate, 30(18), 7529–7553. https://doi.org/10.1175/JCLI-D-16-0810.1
- Capotondi, A., Deser, C., Phillips, A. S., Okumura, Y., & Larson, S. M. (2020). ENSO and Pacific decadal variability in the community Earth system model version 2. Journal of Advances in Modeling Earth Systems, 12(12), e2019MS002022. https://doi.org/10.1029/2019MS002022
- Chylek, P., Klett, J. D., Lesins, G., Dubey, M. K., & Hengartner, N. (2014). The Atlantic Multidecadal Oscillation as a dominant factor of oceanic influence on climate. *Geophysical Research Letters*, 41(5), 1689–1697. https://doi.org/10.1002/2014GL059274
- Clement, A., Bellomo, K., Murphy, L. N., Cane, M. A., Mauritsen, T., R\u00e4del, G., & Stevens, B. (2015). The Atlantic Multidecadal Oscillation without a role for ocean circulation. *Science*, 350(6258), 320–324. https://doi.org/10.1126/science.aab3980
- Dai, Z., Wang, B., Zhu, L., Liu, J., Sun, W., Li, L., et al. (2022). Atlantic multidecadal variability response to external forcing during the past two millennia. Journal of Climate, 35(24), 8103–8115. https://doi.org/10.1175/JCLI-D-21-0986.1
- Delworth, T. L., & Mann, M. E. (2000). Observed and simulated multidecadal variability in the Northern Hemisphere. *Climate Dynamics*, 16(9), 661–676. https://doi.org/10.1007/s003820000075
- Delworth, T. L., Zeng, F., Zhang, L., Zhang, R., Vecchi, G. A., & Yang, X. (2017). The central role of ocean dynamics in connecting the North Atlantic oscillation to the extratropical component of the Atlantic multidecadal oscillation. *Journal of Climate*, 30(10), 3789–3805. https://doi. org/10.1175/JCLI-D-16-0358.1
- Deser, C., Phillips, A., Bourdette, V., & Teng, H. (2012). Uncertainty in climate change projections: The role of internal variability. *Climate Dynamics*, 38(3), 527–546. https://doi.org/10.1007/s00382-010-0977-x
- Deser, C., & Phillips, A. S. (2021). Defining the internal component of Atlantic multidecadal variability in a changing climate. *Geophysical Research Letters*, 48(22), e2021GL095023. https://doi.org/10.1029/2021GL095023
- Deser, C., & Phillips, A. S. (2023). Spurious Indo-Pacific connections to internal Atlantic multidecadal variability introduced by the global temperature residual method. *Geophysical Research Letters*, 50(3), e2022GL100574. https://doi.org/10.1029/2022GL100574
- Enfield, D. B., Mestas-Nuñez, A. M., & Trimble, P. J. (2001). The Atlantic multidecadal oscillation and its relation to rainfall and river flows in the continental U.S. *Geophysical Research Letters*, 28(10), 2077–2080. https://doi.org/10.1029/2000GL012745
- Fang, S.-W., Khodri, M., Timmreck, C., Zanchettin, D., & Jungclaus, J. (2021). Disentangling internal and external contributions to Atlantic multidecadal variability over the past millennium. *Geophysical Research Letters*, 48(23), e2021GL095990. https://doi.org/10.1029/ 2021GL095990
- Fernandez (2024). afernandez74/MTM-SVD: CESM LME MTM-SVD analysis and result visualization code (1.0). Zenodo. https://doi.org/10. 5281/zenodo.14036922
- Folland, C. K., Palmer, T. N., & Parker, D. E. (1986). Sahel rainfall and worldwide sea temperatures, 1901–85. Nature, 320(6063), 602–607. https://doi.org/10.1038/320602a0
- Frankcombe, L. M., England, M. H., Kajtar, J. B., Mann, M. E., & Steinman, B. A. (2018). On the choice of ensemble mean for estimating the forced signal in the presence of internal variability. *Journal of Climate*, 31(14), 5681–5693. https://doi.org/10.1175/JCLI-D-17-0662.1
- Frankignoul, C., Gastineau, G., & Kwon, Y.-O. (2017). Estimation of the SST response to anthropogenic and external forcing and its impact on the Atlantic multidecadal oscillation and the Pacific decadal oscillation. *Journal of Climate*, 30(24), 9871–9895. https://doi.org/10.1175/JCLI-D-17-0009.1

# 19448007, 2025, 11, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2024GL113393, Wiley Online Library on [02/06/2025]. See the Terms and Conditions (https:// conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creativ

- Fyfe, J. C., Kharin, V. V., Santer, B. D., Cole, J. N. S., & Gillett, N. P. (2021). Significant impact of forcing uncertainty in a large ensemble of climate model simulations. *Proceedings of the National Academy of Sciences*, 118(23), e2016549118. https://doi.org/10.1073/pnas. 2016549118
- Gao, C., Robock, A., & Ammann, C. (2008). Volcanic forcing of climate over the past 1500 years: An improved ice core-based index for climate models. Journal of Geophysical Research, 113(D23), D23111. https://doi.org/10.1029/2008JD010239
- Goldenberg, S. B., Landsea, C. W., Mestas-Nuñez, A. M., & Gray, W. M. (2001). The recent increase in Atlantic hurricane activity: Causes and implications. Science, 293(5529), 474–479. https://doi.org/10.1126/science.1060040
- Hawkins, E., & Sutton, R. T. (2009). The potential to narrow uncertainty in regional climate predictions. Bulletin of the American Meteorological Society, 90(8), 1095–1108. https://doi.org/10.1175/2009BAMS2607.1
- Heidemann, H., Cowan, T., Power, S. B., & Henley, B. J. (2024). Statistical relationships between the interdecadal Pacific oscillation and El Niño– Southern Oscillation. *Climate Dynamics*, 62(3), 2499–2515. https://doi.org/10.1007/s00382-023-07035-8
- Hua, W., Dai, A., & Qin, M. (2018). Contributions of internal variability and external forcing to the recent Pacific decadal variations. *Geophysical Research Letters*, 45(14), 7084–7092. https://doi.org/10.1029/2018GL079033
- Hurrell, J. W., Hack, J. J., Shea, D., Caron, J. M., & Rosinski, J. (2008). A new sea surface temperature and sea ice boundary dataset for the community atmosphere model. *Journal of Climate*, 21(19), 5145–5153. https://doi.org/10.1175/2008JCLI2292.1
- Kerr, R. A. (2000). A North Atlantic climate pacemaker for the centuries. Science, 288(5473), 1984–1985. https://doi.org/10.1126/science.288. 5473.1984
- Kim, W. M., Yeager, S., & Danabasoglu, G. (2020). Atlantic multidecadal variability and associated climate impacts initiated by ocean thermohaline dynamics. *Journal of Climate*, 33(4), 1317–1334. https://doi.org/10.1175/JCLI-D-19-0530.1
- Klavans, J. M., Clement, A. C., Cane, M. A., & Murphy, L. N. (2022). The evolving role of external forcing in North Atlantic SST variability over the last millennium. *Journal of Climate*, 35(9), 2741–2754. https://doi.org/10.1175/JCLI-D-21-0338.1
- Knight, J. R. (2009). The Atlantic multidecadal oscillation inferred from the forced climate response in coupled general circulation models. Journal of Climate, 22(7), 1610–1625. https://doi.org/10.1175/2008JCLI2628.1
- Knight, J. R., Allan, R. J., Folland, C. K., Vellinga, M., & Mann, M. E. (2005). A signature of persistent natural thermohaline circulation cycles in observed climate. *Geophysical Research Letters*, 32(20), L20708. https://doi.org/10.1029/2005GL024233
- Knight, J. R., Folland, C. K., & Scaife, A. A. (2006). Climate impacts of the Atlantic multidecadal oscillation. *Geophysical Research Letters*, 33(17), L17706. https://doi.org/10.1029/2006GL026242
- Kravtsov, S., & Callicutt, D. (2017). On semi-empirical decomposition of multidecadal climate variability into forced and internally generated components. *International Journal of Climatology*, 37(12), 4417–4433. https://doi.org/10.1002/joc.5096
- Lapointe, F., Bradley, R. S., Francus, P., Balascio, N. L., Abbott, M. B., Stoner, J. S., et al. (2020). Annually resolved Atlantic sea surface temperature variability over the past 2,900 y. Proceedings of the National Academy of Sciences of the United States of America, 117(44), 27171–27178. https://doi.org/10.1073/pnas.2014166117
- Lehner, F., Deser, C., Maher, N., Marotzke, J., Fischer, E. M., Brunner, L., et al. (2020). Partitioning climate projection uncertainty with multiple large ensembles and CMIP5/6. *Earth System Dynamics*, 11(2), 491–508. https://doi.org/10.5194/esd-11-491-2020
- Li, Z., Zhang, W., Jin, F.-F., Stuecker, M. F., Sun, C., Levine, A. F. Z., et al. (2020). A robust relationship between multidecadal global warming rate variations and the Atlantic Multidecadal Variability. *Climate Dynamics*, 55(7), 1945–1959. https://doi.org/10.1007/s00382-020-05362-8
- Mann, M. E., & Park, J. (1994). Global-scale modes of surface temperature variability on interannual to century timescales. Journal of Geophysical Research, 99(D12), 25819–25833. https://doi.org/10.1029/94JD02396
- Mann, M. E., & Park, J. (1999). Oscillatory spatiotemporal signal detection in climate studies: A multiple-taper spectral domain approach. In R. Dmowska & B. Saltzman (Eds.), Advances in geophysics (Vol. 41, pp. 1–131). Elsevier. https://doi.org/10.1016/S0065-2687(08)60026-6
- Mann, M. E., Steinman, B. A., Brouillette, D. J., Fernandez, A., & Miller, S. K. (2022). On the estimation of internal climate variability during the preindustrial past millennium. *Geophysical Research Letters*, 49(2), e2021GL096596. https://doi.org/10.1029/2021GL096596
- Mann, M. E., Steinman, B. A., Brouillette, D. J., & Miller, S. K. (2021). Multidecadal climate oscillations during the past millennium driven by volcanic forcing. *Science*, 371(6533), 1014–1019. https://doi.org/10.1126/science.abc5810
- Mann, M. E., Steinman, B. A., & Miller, S. K. (2020). Absence of internal multidecadal and interdecadal oscillations in climate model simulations. *Nature Communications*, 11(1), 49. https://doi.org/10.1038/s41467-019-13823-w
- Mavilia, I., Bellucci, A., J. Athanasiadis, P., Gualdi, S., Msadek, R., & Ruprich-Robert, Y. (2018). On the spectral characteristics of the Atlantic multidecadal variability in an ensemble of multi-century simulations. *Climate Dynamics*, 51(9), 3507–3520. https://doi.org/10.1007/s00382-018-4093-7
- McCarthy, G. D., Haigh, I. D., Hirschi, J. J.-M., Grist, J. P., & Smeed, D. A. (2015). Ocean impact on decadal Atlantic climate variability revealed by sea-level observations. *Nature*, 521(7553), 508–510. https://doi.org/10.1038/nature14491
- McGregor, S., Khodri, M., Maher, N., Ohba, M., Pausata, F. S. R., & Stevenson, S. (2020). The effect of strong volcanic eruptions on ENSO. In *El Niño Southern Oscillation in a changing climate* (pp. 267–287). American Geophysical Union (AGU). https://doi.org/10.1002/ 9781119548164.ch12
- Meehl, G. A., Hu, A., Castruccio, F., England, M. H., Bates, S. C., Danabasoglu, G., et al. (2021). Atlantic and Pacific tropics connected by mutually interactive decadal-timescale processes. *Nature Geoscience*, 14(1), 36–42. https://doi.org/10.1038/s41561-020-00669-x

Miao, J., Ge, Y., & Jiang, D. (2024). Changes of Atlantic multidecadal variability during the medieval climate anomaly and little ice age and associated causes. *Journal of Geophysical Research: Atmospheres*, 129(1), e2023JD039966. https://doi.org/10.1029/2023JD039966

- Michel, S. L. L., Swingedouw, D., Ortega, P., Gastineau, G., Mignot, J., McCarthy, G., & Khodri, M. (2022). Early warning signal for a tipping point suggested by a millennial Atlantic multidecadal variability reconstruction. *Nature Communications*, 13(1), 5176. https://doi.org/10.1038/ s41467-022-32704-3
- Miller, G. H., Geirsdóttir, Á., Zhong, Y., Larsen, D. J., Otto-Bliesner, B. L., Holland, M. M., et al. (2012). Abrupt onset of the little ice age triggered by volcanism and sustained by sea-ice/ocean feedbacks. *Geophysical Research Letters*, 39(2), L02708. https://doi.org/10.1029/ 2011GL050168
- Murphy, L. N., Bellomo, K., Cane, M., & Clement, A. (2017). The role of historical forcings in simulating the observed Atlantic multidecadal oscillation. *Geophysical Research Letters*, 44(5), 2472–2480. https://doi.org/10.1002/2016GL071337
- Newman, M., Alexander, M. A., Ault, T. R., Cobb, K. M., Deser, C., Lorenzo, E. D., et al. (2016). The Pacific decadal oscillation, revisited. *Journal of Climate*, 29(12), 4399–4427. https://doi.org/10.1175/JCLI-D-15-0508.1
- O'Reilly, C. H., Zanna, L., & Woollings, T. (2019). Assessing external and internal sources of Atlantic multidecadal variability using models, proxy data, and early instrumental indices. *Journal of Climate*, 32(22), 7727–7745. https://doi.org/10.1175/JCLI-D-19-0177.1
- Otterå, O. H., Bentsen, M., Drange, H., & Suo, L. (2010). External forcing as a metronome for Atlantic multidecadal variability. *Nature Geoscience*, *3*(10), 688–694. https://doi.org/10.1038/ngeo955

- Otto-Bliesner, B. L., Brady, E. C., Fasullo, J., Jahn, A., Landrum, L., Stevenson, S., et al. (2016). Climate variability and change since 850 CE: An ensemble approach with the community Earth system model. *Bulletin of the American Meteorological Society*, 97(5), 735–754. https://doi.org/ 10.1175/BAMS-D-14-00233.1
- Pausata, F. S. R., Zhao, Y., Zanchettin, D., Caballero, R., & Battisti, D. S. (2023). Revisiting the mechanisms of ENSO response to tropical volcanic eruptions. *Geophysical Research Letters*, 50(3), e2022GL102183. https://doi.org/10.1029/2022GL102183
- Qin, M., Dai, A., & Hua, W. (2020). Quantifying contributions of internal variability and external forcing to Atlantic multidecadal variability since 1870. Geophysical Research Letters, 47(22), e2020GL089504. https://doi.org/10.1029/2020GL089504
- Robson, J., Sutton, R. T., Menary, M. B., & Lai, M. W. K. (2023). Contrasting internally and externally generated Atlantic multidecadal variability and the role for AMOC in CMIP6 historical simulations. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 381(2262), 20220194. https://doi.org/10.1098/rsta.2022.0194
- Ruprich-Robert, Y., Msadek, R., Castruccio, F., Yeager, S., Delworth, T., & Danabasoglu, G. (2017). Assessing the climate impacts of the observed Atlantic multidecadal variability using the GFDL CM2.1 and NCAR CESM1 global coupled models. *Journal of Climate*, 30(8), 2785–2810. https://doi.org/10.1175/JCLI-D-16-0127.1
- Sanchez, S. C., Amaya, D. J., Miller, A. J., Xie, S.-P., & Charles, C. D. (2019). The Pacific meridional mode over the last millennium. *Climate Dynamics*, 53(5), 3547–3560. https://doi.org/10.1007/s00382-019-04740-1
- Schlesinger, M. E., & Ramankutty, N. (1994). An oscillation in the global climate system of period 65–70 years. *Nature*, 367(6465), 723–726. https://doi.org/10.1038/367723a0
- Semenov, V. A., Latif, M., Dommenget, D., Keenlyside, N. S., Strehz, A., Martin, T., & Park, W. (2010). The impact of North Atlantic–Arctic multidecadal variability on Northern Hemisphere surface air temperature. *Journal of Climate*, 23(21), 5668–5677. https://doi.org/10.1175/ 2010JCL13347.1
- Slawinska, J., & Robock, A. (2018). Impact of volcanic eruptions on decadal to centennial fluctuations of Arctic sea ice extent during the last millennium and on initiation of the little ice age. Journal of Climate, 31(6), 2145–2167. https://doi.org/10.1175/JCLI-D-16-0498.1
- Steinman, B. A., Mann, M. E., & Miller, S. K. (2015). Atlantic and Pacific multidecadal oscillations and Northern Hemisphere temperatures. Science, 347(6225), 988–991. https://doi.org/10.1126/science.1257856
- Sutton, R. T., & Hodson, D. L. R. (2005). Atlantic Ocean forcing of North American and European summer climate. *Science*, 309(5731), 115–118. https://doi.org/10.1126/science.1109496
- Sutton, R. T., McCarthy, G. D., Robson, J., Sinha, B., Archibald, A. T., & Gray, L. J. (2018). Atlantic multidecadal variability and the U.K. ACSIS Program. Bulletin of the American Meteorological Society, 99(2), 415–425. https://doi.org/10.1175/BAMS-D-16-0266.1
- Takahashi, C., & Watanabe, M. (2016). Pacific trade winds accelerated by aerosol forcing over the past two decades. *Nature Climate Change*, 6(8), 768–772. https://doi.org/10.1038/nclimate2996
- Thompson, D. W. J., Barnes, E. A., Deser, C., Foust, W. E., & Phillips, A. S. (2015). Quantifying the role of internal climate variability in future climate trends. *Journal of Climate*, 28(16), 6443–6456. https://doi.org/10.1175/JCLI-D-14-00830.1
- Thomson, D. J. (1982). Spectrum estimation and harmonic analysis. Proceedings of the IEEE, 70(9), 1055–1096. https://doi.org/10.1109/PROC. 1982.12433
- Ting, M., Kossin, J. P., Camargo, S. J., & Li, C. (2019). Past and future hurricane intensity change along the U.S. East coast. Scientific Reports, 9(1), 7795. https://doi.org/10.1038/s41598-019-44252-w
- Ting, M., Kushnir, Y., Seager, R., & Li, C. (2009). Forced and internal twentieth-century SST trends in the North Atlantic. Journal of Climate, 22(6), 1469–1481. https://doi.org/10.1175/2008JCL12561.1
- Torrence, C., & Compo, G. P. (1998). A practical guide to wavelet analysis. Bulletin of the American Meteorological Society, 79(1), 61–78. https:// doi.org/10.1175/1520-0477(1998)079<0061:APGTWA>2.0.CO;2
- Trenberth, K. E., & Shea, D. J. (2006). Atlantic hurricanes and natural variability in 2005. *Geophysical Research Letters*, 33(12), L12704. https://doi.org/10.1029/2006GL026894
- Wang, J., Yang, B., Ljungqvist, F. C., Luterbacher, J., Osborn, T. J., Briffa, K. R., & Zorita, E. (2017). Internal and external forcing of multidecadal Atlantic climate variability over the past 1,200 years. *Nature Geoscience*, 10(7), 512–517. https://doi.org/10.1038/ngeo2962
- Watanabe, M., & Tatebe, H. (2019). Reconciling roles of sulphate aerosol forcing and internal variability in Atlantic multidecadal climate changes. Climate Dynamics, 53(7), 4651–4665. https://doi.org/10.1007/s00382-019-04811-3
- Wills, R. C. J., Battisti, D. S., Armour, K. C., Schneider, T., & Deser, C. (2020). Pattern recognition methods to separate forced responses from internal variability in climate model ensembles and observations. *Journal of Climate*, 33(20), 8693–8719. https://doi.org/10.1175/JCLI-D-19-0855.1
- Xu, H., Wang, T., & Wang, H. (2024). The Interdecadal Pacific Oscillation is responsible for the linkage of decadal changes in precipitation and moisture in arid central Asia and the humid Asian monsoon region during the last millennium. *Climate of the Past*, 20(1), 107–119. https://doi. org/10.5194/cp-20-107-2024
- Yang, Y.-M., An, S.-I., Wang, B., & Park, J. H. (2020). A global-scale multidecadal variability driven by Atlantic multidecadal oscillation. *National Science Review*, 7(7), 1190–1197. https://doi.org/10.1093/nsr/nwz216
- Yeager, S. (2020). The abyssal origins of North Atlantic decadal predictability. Climate Dynamics, 55(7), 2253-2271. https://doi.org/10.1007/s00382-020-05382-4
- Zhang, L., & Zhao, C. (2015). Processes and mechanisms for the model SST biases in the North Atlantic and North Pacific: A link with the Atlantic meridional overturning circulation. *Journal of Advances in Modeling Earth Systems*, 7(2), 739–758. https://doi.org/10.1002/2014MS000415
- Zhang, R. (2017). On the persistence and coherence of subpolar sea surface temperature and salinity anomalies associated with the Atlantic multidecadal variability. *Geophysical Research Letters*, 44(15), 7865–7875. https://doi.org/10.1002/2017GL074342
- Zhang, R., & Delworth, T. L. (2006). Impact of Atlantic multidecadal oscillations on India/Sahel rainfall and Atlantic hurricanes. Geophysical Research Letters, 33(17), L17712. https://doi.org/10.1029/2006GL026267
- Zhang, R., & Delworth, T. L. (2007). Impact of the Atlantic multidecadal oscillation on North Pacific climate variability. Geophysical Research Letters, 34(23). https://doi.org/10.1029/2007GL031601
- Zhou, X., Lang, X., & Jiang, D. (2020). Teleconnections between the Atlantic multidecadal oscillation and eastern China summer precipitation during the medieval climate anomaly and little ice age. *The Holocene*, 30(12), 1694–1705. https://doi.org/10.1177/0959683620950413

#### **References From the Supporting Information**

- Correa-Ramirez, M., & Hormazabal, S. (2012). MultiTaper Method-Singular Value Decomposition (MTM-SVD): variabilidad espaciofrecuencia de las fluctuaciones del nivel del mar en el Pacífico suroriental. Latin American Journal of Aquatic Research, 40(4), 1039– 1060. https://doi.org/10.3856/vol40-issue4-fulltext-19
- Khider, D., Emile-Geay, J., Zhu, F., James, A., Landers, J., Ratnakar, V., & Gil, Y. (2022). Pyleoclim: Paleoclimate timeseries analysis and visualization with Python. Paleoceanography and Paleoclimatology, 37(10), e2022PA004509. https://doi.org/10.1029/2022PA004509
- Liu, W., Fedorov, A. V., Xie, S.-P., & Hu, S. (2020). Climate impacts of a weakened Atlantic meridional overturning circulation in a warming climate. *Science Advances*, 6(26). https://doi.org/10.1126/sciadv.aaz4876
- Slepian, D., & Pollak, H. O. (1961). Prolate spheroidal wave functions, Fourier analysis and uncertainty I. *Bell System Technical Journal*, 40(1), 43–63. https://doi.org/10.1002/j.1538-7305.1961.tb03976.x